Author’s response to reviews

Title: A prospective cohort study of the effect of different methods of drainage of the neuropathic bladder on the occurrence of symptomatic urinary infection, and adverse events related to the urinary drainage system in spinal cord injury patients: a research protocol

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Bhakul M Soni (Bhakul.Soni@mail.soh-tr.nwest.nhs.uk)
Singh Gurpreet (Gurpreet.Singh@mail.soh-tr.nwest.nhs.uk)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Tun Oo (T.Oo@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)
Keith F Parsons (Keith.Parsons@hotmail.com)

Version: 2 Date: 13 Oct 2001

Dear Editor
We are grateful to you and to the reviewer for the valuable comments. We revised the manuscript to incorporate the suggestions made by the reviewer.
The manuscript has been presented according to BMC instructions.
A structured abstract has been used.
Typographic errors have been corrected.
A description of competing interest has been given.
Nature of sponsorship is stated.
The entire manuscript has been revised as per the comments made by the reviewer.
We shall be happy to make further changes if you desire so.
Many thanks for considering our submission
Yours sincerely
Dr S Vaidyanathan
REgional Spinal Injuries Centre
District General Hospital
Town Lane
Southport
Merseyside PR8 6PN
United Kingdom
Telephone: + 44 1704 547471 Pager 671
Fax: + 44 1704 543156
E-mail: vaidyanathansiu@hotmail.com